Author's response to reviews

Title: A study of the provision of hospital based dental General Anaesthetic services for children in the North West of England: Part 2. The views and experience of families and dentists regarding service needs, treatment and prevention

Authors:

Michaela Goodwin (michaela.goodwin@manchester.ac.uk)
Iain Pretty (iain.pretty@manchester.ac.uk)
Caroline Sanders (Caroline.Sanders@manchester.ac.uk)

Version: 4 Date: 10 March 2015

Author's response to reviews:

We are very grateful for the time and effort that the reviewers have made in helping us improve this work.

In response to the editors comment we have updated line 209 to describe the informed written consent they parents gave both for themselves and their child

For table 1 and 2 we have removed gender for both the child and dentist and for the children we have provided an age bracket rather than actual age. If whether the mother or father also needs to be removed the authors would be happy to also make this amendment.

I hope that this is satisfactory. Do please contact me if I can provide anything further.

Kindest Regards

Michaela Goodwin